Author's response to reviews

Title: Contribution of MRI in the diagnosis of talus skip metastases of calcaneus Ewing’s sarcoma in child: a case report

Authors:

Hicham Jalal (jhicham2000@hotmail.com)
Zoubida Belhadj (belhajz@yahoo.fr)
Hind Enneddam (enneddam-hind@hotmail.com)
Mohammed Madhar (madmed24@yahoo.fr)
Tarik Fikry (tfikry@menara.ma)
Omar Essadki (o.essadki@gmail.com)
Ahmed Ousehal (a.ousehal@voila.fr)

Version: 2 Date: 27 March 2011

Author's response to reviews:

The authors followed the recommendations of the authors and have introduced the following changes to their manuscript:

- correction of grammatical errors
- added missing paragraphs: Abbreviations, Authors' contributions, Acknowledgements
- format references and consent section

And we change the font size according to the instructions for authors of JMCRI.